

REVIEW HISTORY

Paper: **“Modeling Exchange Rate Volatility in the Presence of Serial Correlations”**

Corresponding Author: Emmanuel Dodzi Kutor Havi

Email: ehavi@mucg.edu.gh

Doi: 10.19044/elp.v6no4a2

Peer review:

Reviewer 1: Blinded

Reviewer 2: Blinded

Published: 29.12.2019

